Dear Sir:

Please find enclosed a new version of our manuscript. We have revised and updated the text and revised the English language, according the reviewer’s suggestion. This version has been approved by all the authors. They have given necessary attention to ensure the integrity of the work. If there are any questions regarding to our submittal, please do not hesitate to contact me: Dr. José A. Girón González, Servicio de Medicina Interna, Hospital Universitario Puerta del Mar, Avda. Ana de Viya 21, 11009 Cádiz, Spain; telephone / fax +34-956-002879, e-mail joseantonio.giron@uca.es.

We appreciate the constructive comments and feel that the changes that have been made in the paper have significantly strengthened it.

Yours sincerely,

José A Girón-González
Reviewer.

This is a well-written review regarding hepatic and extrahepatic benefits after achieving HCV cure. However, there is much literature on this topic and therefore much of the written concepts are redundant.

Answer.

Redundances in the text have been suppressed.